Accelerator Status Report

Run Schedule: March 11, 2010 to July 22, 2010
May 5, 2010
Injector    Energy 56.62 MeV
· (5/4)Some work on the Poekel cell was done to reduce the “ringing”.

· An injector study showed that the Wien filers may be the cause of the 5-second beam motion. The investigation is not complete yet.
· During yesterday’s beam studies it was found that the voltage was not stable across each helicity state of the Pockel cell. This might the cause of the Hall A helicity correlated asymmetry issues.

Accelerator   Linac Energy 503.25 MeV
· (5/3) The accelerator optics was optimized, and Hall B and BSY were changed to pass 5. 

· (5/2) The CHL tripped possibly due to a power glitch, which caused more than five hours of downtime. 

· (5/1) After the open house the accelerator was recovered nicely.

· (4/30) Beam stopped for the open house.

· The accelerator has been stable.

· Zones 1L18 and 2L21 were recovered.

· Cavity 0L03-1 mismatched by 40% but did not on the RF commander screen. The mismatch caused ep ion chamber trips. The problem was fixed by replacing the control module.
· Beam restored to the BSY dump and the vacuum is still good. 

Hall A---Pass 1, Prex from March 11 to May 24, 2010
· (5/2) After the open house Hall A started to take straight-ahead mode beam due to the work on the Compton laser cavity mirrors.  

· (4/29) Hall A stopped taking beam to fix a vacuum leak downstream of the target chamber. 

· Hall A radiation does rates are higher than 1R/h, which requires a watch when in access.

· Hall A target chamber vacuum is not stable. We should pay attention

· Hall A has helicity-correlated asymmetry issues. During yesterday’s beam studies it was found that the voltage was not stable across each helicity state of the Pockel cell. This might the cause of the helicity correlated asymmetry issues.

· The experts are investigating.

· Hall A Compton signal to noise ratio was getting bad, which needed some fine-tuning of beam. Yesterday it was improved by tuning the injector and Hall A line.
· Hall A Moller target current limits updated. 5uA for standard runs. 15uA for the special beam mode runs. M. Joyce will modify the Maxjuice.

Hall B ---Pass 5, g9-Frost from March 11 to July 22, 2010
· (5/3) Hall B was changed to pass 5.

· No problems with Hall B beam

Hall C---QWeak from May 25 to July 22, 2010
· QWeak installation
